Author's response to reviews

Title: Improving motivation among primary health care workers in Tanzania: a health worker perspective

Authors:
Rachel N Manongi (rmanongi@yahoo.co.uk)
Tanya C Marchant (tanya.marchant@lshtm.ac.uk)
Christian I Bygbjerg (i.bygbjerg@pubhealth.ku.dk)

Version: 3 Date: 27 October 2005

Author's response to reviews:

Reviewer's comments

1. Thanks

2. Thanks. See additional section in the text. This was an explorative study and not a comparative one which was describing motivation among health workers. The government health facilities were chosen because the govt. health system is the back-bone of the Tanzanians' health system, and they have majority of health workers compared to private sector (see the text).

3. Thanks. Explanation has been added in the text. (1st author is fluent in English and Swahili, so an intermediate interpreter was not needed), but on the other hand, we are aware of the pitfalls of translating from one language to another. Ethics section changed sees the text.

4. Thanks see current Tables 1 and 2.

5. Thanks

6. Added see the abstract

7. Thanks